Author's response to reviews

Title: Short-term outcome of patients with possible transient ischemic attacks: a prospective study

Authors:

Mariana Correia (mariana.d.correia@gmail.com)
Ana C Fonseca (catarinagfonseca@gmail.com)
Patrícia Canhão (pcanhao@fm.ul.pt)

Version: 4 Date: 8 April 2015

Author's response to reviews: see over
Dear Editor,

We made the following changes in the manuscript:

1. Please clarify in the manuscript if consent was oral or written - we clarified that the consent was oral

   *Enrolled patients gave oral informed consent.*

2. Please include Abbreviations section - *We included the Abbreviations section*

   **Abbreviations**
   
   *CT: Computed tomography*
   
   *DWI: Diffusion-weighted image*
   
   *ECG: Electrocardiogram*
   
   *ED: Emergency department*
   
   *MRI: Magnetic resonance imaging*
   
   *MI: Myocardial infarction*
   
   *TIA: Transient ischemic attack*

3. Please include acknowledgements section with details of any funding included- we added the sources of funding.

   *This study was supported by Instituto de Medicina Molecular of Lisbon*

Sincerely,

Patrícia Canhão